Author's response to reviews

Title: Healthy Hearts - a community-based primary prevention programme to reduce coronary heart disease

Authors:

Gill Richardson (Gill.Richardson@nphs.wales.nhs.uk)
Hugo van Woerden (h.vanwoerden@sgul.ac.uk)
Lucy Morgan (Lucy.Morgan@nphs.wales.nhs.uk)
Rhiannon Edwards (Rhiannon.Edwards@nphs.wales.nhs.uk)
Monica Harries (Monica.Harries@nphs.wales.nhs.uk)
Elaine Hancock (ehancock@sky.com)
Susan Sroczynska (Susan.Sroczynska@nphs.wales.nhs.uk)
Mererid Bowley (mererid.bowley@nphs.wales.nhs.uk)

Version: 2 Date: 5 September 2007

Author's response to reviews:

Minor amendments have been made to the authors, the acknowledgements section and the length of string used in the study.